ORIGINAL RESEARCH



# Impact of Crisaborole on Sleep Outcomes in Pediatric Patients with Mild-to-Moderate Atopic Dermatitis

Joseph Fowler  $\cdot$  Jeffrey Sugarman  $\cdot$  Lawrence Sher  $\cdot$  Chuanbo Zang  $\cdot$ 

John L. Werth · Daniela E. Myers · Daniela Graham · Alexander Agyei Marfo 💿 · Liza Takiya

Received: October 26, 2022 / Accepted: January 31, 2023 / Published online: February 22, 2023 © The Author(s) 2023

### ABSTRACT

*Introduction*: Using data from three clinical trials, the effect of crisaborole treatment on sleep outcomes for pediatric patients with atopic dermatitis (AD) and their families was examined.

*Methods*: This analysis comprised patients aged 2 to < 16 years from the double-blind phase 3 CrisADe CORE 1 (NCT02118766) and CORE 2 (NCT02118792) studies, families of patients aged 2 to < 18 years from CORE 1 and CORE 2, and patients aged 3 months to < 2 years from the open-label phase 4 CrisADe CARE 1 study

J. Fowler

Dermatology Specialists Research LLC and University of Louisville School of Medicine, Louisville, KY, USA

J. Sugarman University of California, San Francisco, CA, USA

L. Sher Palos Verdes Medical Group, Rolling Hills Estates, CA, USA

C. Zang  $\cdot$  J. L. Werth  $\cdot$  D. E. Myers  $\cdot$  L. Takiya Pfizer Inc., Collegeville, PA, USA

D. Graham Pfizer Inc., Groton, CT, USA

A. A. Marfo (⊠) Pfizer Inc., 235 East 42Nd Street, 235:13/S33, New York, NY 10017, USA e-mail: alexander.a.marfo@pfizer.com (NCT03356977), all with mild-to-moderate AD who received crisaborole ointment 2% twice daily for 28 days. Sleep outcomes were assessed via the Children's Dermatology Life Quality Index and Dermatitis Family Impact questionnaires in CORE 1 and CORE 2 and the Patient-Oriented Eczema Measure questionnaire in CARE 1.

**Results:** In CORE 1 and CORE 2, a significantly lower proportion of crisaborole-treated patients than vehicle-treated patients reported sleep disruption at day 29 (48.5% versus 57.7%, p = 0.001). The proportion of families whose sleep was affected by their child's AD in the preceding week was also significantly lower in the crisaborole group (35.8% versus 43.1%, p = 0.02) at day 29. At day 29 in CARE 1, the proportion of crisaborole-treated patients who experienced  $\geq$  1 night of disturbed sleep in the previous week decreased by 32.1% from baseline.

*Conclusion*: These results suggest that crisaborole improves sleep outcomes in pediatric patients with mild-to-moderate AD and their families.

# PLAIN LANGUAGE SUMMARY

Atopic dermatitis (AD), also known as eczema, is a chronic skin disease that causes red or flaky skin patches that can become infected and itch.

Children with AD often experience sleep disturbance, including difficulty falling asleep, restless sleep, waking up more frequently, and daytime drowsiness. Problems with sleep quality negatively impact children with AD, as well as their caregivers. Crisaborole ointment is applied to the skin and has been shown to improve the symptoms of AD in children and adults. This study examined how treatment with crisaborole affected sleep quality for children and their caregivers in three clinical trials. Children in these studies took crisaborole for 28 days. Researchers found that crisaborole treatment improved sleep in children with mild-to-moderate AD and their caregivers. This was determined using four measures. First, a smaller proportion of children who were treated with crisaborole experienced sleep disruption compared with those to whom a vehicle was applied (an ointment with no drug). Second, a smaller proportion of caregivers of children with AD who were treated with crisaborole reported effects on their sleep, compared with children to whom a vehicle was applied. Third, a smaller proportion of children with AD who were treated with crisaborole, as well as their caregivers, had  $\geq 1$  night per week of disturbed sleep after treatment compared with before treatment. Fourth, the caregivers of children treated with crisaborole reported significantly less exhaustion and tiredness because of the child's AD. These results suggest that treatment with crisaborole improves sleep outcomes in children with mild-to-moderate AD and their caregivers.

**Keywords:** Atopic dermatitis; Crisaborole; Pediatrics; Sleep

### **Key Summary Points**

#### Why carry out this study?

Pediatric patients with atopic dermatitis (AD) may experience sleep disturbances, which may lead to daytime fatigue and impaired quality of life for patients and their families.

This post hoc analysis examined how crisaborole ointment 2%, an antiinflammatory nonsteroidal phosphodiesterase 4 inhibitor, affected sleep outcomes for pediatric patients and their families across the CORE 1, CORE 2, and CARE 1 studies.

#### What was learned from the study?

After 28 days of twice-daily use, crisaborole treatment resulted in a lower proportion of patients reporting sleep disruption compared with vehicle, a lower proportion of families whose sleep was affected compared with vehicle, and a reduction from baseline in the proportion of patients with  $\geq 1$  night per week of disturbed sleep.

These results suggest that treatment with crisaborole improves sleep outcomes in pediatric patients with mild-to-moderate AD and their families.

# INTRODUCTION

Atopic dermatitis (AD) is associated with a significant psychosocial burden that affects patients, their caregivers, and society [1, 2]. The estimated prevalence of sleep disturbance in US pediatric patients aged 5-17 years with AD is approximately 67% [3]. Sleep disturbances experienced by patients with AD can take many forms, including difficulty in initiating sleep, restless sleep, increased waking, and daytime drowsiness [1, 2, 4]. As the severity of AD increases, so does the magnitude of those negative effects on sleep. AD flares also exacerbate negative impact on sleep [2, 5]. A USA-based survey of more than 90,000 children aged 0-17 years revealed that children who reported having AD within the past year had a higher chance of experiencing impaired sleep, and children with severe AD reported more instances of sleep disruption than children with mild or moderate AD [5, 6]. Similarly, a cohort study almost 14,000 British children aged of

2–16 years found that children with AD, even those with cases of mild or inactive disease, had significantly worse sleep quality relative to children without AD [7].

Problems with sleep quality appear to have a substantial impact on the quality of life (QoL) of patients with AD, even among patients in clinical remission. Children with AD often exhibit behavioral issues that may be linked to sleep disturbances, including negative self-esteem, poor performance in school, and school absenteeism [1, 8–11]. Poor sleep experienced by children with AD has also been linked to increased stress on their families, thereby imposing a psychosocial burden on the family as a whole [1, 2, 12]. A longitudinal study in the UK of almost 12,000 mother-child pairs found that mothers of children with AD experienced more sleep issues than mothers of children without AD, including difficulty falling asleep, insufficient sleep, and daytime exhaustion. Effects on family members' sleep could have negative effects on many aspects of their QoL, for example, performance in the workplace and coping skills at work and home have been observed to significantly decrease after a loss of as little as 1 or 2 h of sleep per night [7].

Crisaborole ointment 2% is a nonsteroidal phosphodiesterase 4 (PDE4) inhibitor, approved in many countries for the treatment of patients aged  $\geq$  3 months with mild-to-moderate AD [13, 14]. US Food and Drug Administration approval of crisaborole was based on the efficacy and safety demonstrated in two identically designed, randomized, double-blind, vehiclecontrolled, 28-day phase 3 studies [CrisADe CORE 1 (NCT02118766) and CrisADe CORE 2 (NCT02118792)] that comprised patients aged  $\geq 2$  years [15]. In both studies, patients treated with crisaborole experienced a significantly greater reduction in AD severity versus those receiving vehicle [15]. In addition, patients treated with crisaborole achieved a significantly greater improvement in pruritus than vehicle-treated patients [15, 16]. Crisaborole was found to be generally safe and well tolerated [15]. The safety of crisaborole was further investigated in infants with mild-tomoderate AD, aged 3 months to < 2 years, in CrisADe CARE 1 (NCT03356977), a multicenter, open-label, single-arm, 28-day phase 4 study. In this study, treatment with crisaborole resulted in improvements in AD severity and reductions in "the number of days skin had been itchy" during the previous week. Crisaborole was well tolerated by patients in CARE 1, and the safety profile was similar to that observed in the pivotal phase III studies [17].

The objective of this post hoc analysis was to examine how crisaborole treatment affected sleep outcomes for pediatric patients and their families across the CORE 1, CORE 2, and CARE 1 studies.

### **METHODS**

#### **Patients and Treatment**

In CORE 1 and CORE 2, the efficacy and safety of crisaborole were compared with vehicle in patients aged > 2 years with AD per Hanifin and Rajka criteria [18], with mild-to-moderate disease per ISGA, and with percentage of treatable body surface area (%BSA) > 5 (excluding the scalp) [15]. Patients were randomly assigned 2:1 to receive crisaborole ointment 2% or vehicle, with treatment applied twice daily to all AD-affected areas, except the scalp, for 28 days. In CARE 1 (NCT03356977), patients aged 3 months to < 2 years with mild-to-moderate AD and %BSA > 5 (excluding the scalp) were included. Open-label crisaborole was applied twice daily to all AD-affected areas of the body (excluding the scalp) for 28 days, avoiding mucous membranes [17].

#### **Outcomes and Assessments**

In CORE 1 and CORE 2, Children's Dermatology Life Quality Index (CDLQI) and Dermatitis Family Impact (DFI) assessments were completed at baseline and day 29. The CDLQI and DFI are questionnaires that assess QoL in pediatric patients with skin disease, and in their families, respectively. Each contains item(s) that assess the effect of AD on sleep. The CDLQI was completed either by the patients or with parental assistance for patients aged < 4 years [19]. The DFI was completed by one caregiver on behalf of the family.

In CORE 1 and CORE 2, sleep outcomes included the responses from patients aged 2 to < 16 years to item 9 of the CDLQI ("Over the last week, how much has your sleep been affected by your skin problem?"). Possible answers were "very much," "quite a lot," "only a little," and "not at all" [20]. The responses from caregivers of patients aged 2 to < 18 years to item 3 of the DFI ("Over the last week, how much effect has your child having eczema had on the sleep of others in the family?") were also analyzed. Possible answers were "very much," "a lot," "a little," and "not at all" [21]. Finally, tiredness and exhaustion were assessed in the families of patients aged 2 to < 18 years via item 7 of the DFI ("Over the last week, how much effect has your child having eczema had on causing tiredness or exhaustion in your child's parents/carers?"). Possible answers were "very much," "a lot," "a little," and "not at all" [21].

In CARE 1, a proxy version of the Patient-Oriented Eczema Measure (POEM) was completed by the parents or guardians of the patient at baseline and at days 8, 15, and 29 [17]. The CDLQI and DFI were not captured in the CARE 1 study [17]. The POEM is a 7-item questionnaire used for monitoring atopic eczema severity, focusing on the illness as experienced by the patient [22]. Each item is measured on a scale of 0-4, with a total score ranging from 0 to 28. Higher scores indicate greater AD symptom impact. In CARE 1, the sleep outcome was represented by responses by parents or guardians of patients to item 2 of the proxy POEM ("Over the last week, on how many nights has your child's sleep been disturbed because of eczema?") [22]. Possible answers were "no days," "1-2 days," "3-4 days," "5-6 days," and "every day" [22].

The institutional review board at each study site approved the study protocol, and written informed consent was provided by parents or legal guardians. The studies were conducted in accordance with the protocol, local legal and regulatory requirements, and the general principles set forth in the International Ethical Guidelines for Biomedical Research Involving Human Subjects, the International Conference on Harmonisation Guideline for Good Clinical Practice, and the Declaration of Helsinki.

#### **Statistical Analysis**

For this post hoc analysis, sleep outcomes were assessed using pooled data from patients aged 2 to < 16 years, and families of patients aged 2 to < 18 years in CORE 1 and CORE 2. *p* values were derived from a Wilcoxon rank sum test for the comparison of crisaborole versus vehicle. No imputation was performed for missing data; observed cases were used. Analyses were performed using the intention-to-treat population, which consisted of all patients who had been randomly assigned and received study drug. For CARE 1, POEM data were summarized descriptively using the full analysis set of patients, which included any patient receiving  $\geq$  1 dose of crisaborole.

### RESULTS

### **Baseline Characteristics**

A total of 1199 patients (crisaborole 796, vehicle 403) and 1291 families (crisaborole 860, vehicle 431) had CDLQI and DFI sleep data collected, respectively, in CORE 1 and 2. POEM sleep data were collected for 137 infants in CARE 1, all of whom had been treated with crisaborole. The baseline and disease characteristics of CORE 1 and CORE 2 patients aged 2 to < 16 years and patients aged 2 to < 18 years were comparable (Table 1). With regard to demographics, the percentage of white patients in CARE 1 was similar to that in CORE 1 and CORE 2 ( $\sim 60\%$ ). but the overall proportion of female patients was lower (36% versus ~ 53%) (Table 1). With regard to baseline disease characteristics, patients in CARE 1 had notably higher mean and median %BSAs than patients in CORE 1 and CORE 2 (Table 1). Patients in all three studies had comparable proportions of patients with ISGA scores of 2 (mild) and 3 (moderate) (Table 1).

#### Table 1 Baseline patient demographics

	CARE 1 patients aged 3 months to < 2 years	CORE 1 and CORE 2 patients aged 2 to < 16 years		CORE 1 and CORE 2 patients aged 2 to < 18 years	
	Crisaborole $n = 137$	Vehicle <i>n</i> = 412	Crisaborole <i>n</i> = 815	Vehicle n = 439	Crisaborole n = 874
Age, years					
Mean (SD)	1.1 (0.54)	7.8 (4.0)	7.7 (4.1)	8.4 (4.4)	8.3 (4.5)
Median (min, max)	1.1 (0.3, 1.9)	8.0 (2, 15)	7.0 (2, 15)	8.0 (2, 17)	8.0 (2, 17)
Female, n (%)	49 (35.8)	216 (52.4)	433 (53.1)	232 (52.8)	466 (53.3)
White, $n$ (%)	84 (61.3)	255 (61.9)	503 (61.7)	273 (62.2)	536 (61.3)
%BSA					
Mean (SD)	28.1 (22.0)	19.1 (17.6)	19.2 (18.7)	18.5 (17.2)	18.9 (18.6)
Median (min, max)	19.0 (5, 94)	12.0 (5, 90)	12.0 (5, 95)	12.0 (5, 90)	12.0 (5, 95)
ISGA, n (%)					
Mild (2)	52 (38.0)	154 (37.4)	309 (37.9)	167 (38.0)	333 (38.1)
Moderate (3)	84 (61.3)	258 (62.6)	506 (62.1)	272 (62.0)	541 (61.9)

%BSA percentage treatable body surface area, ISGA Investigator's Static Global Assessment, SD standard deviation

#### **Sleep Outcomes**

To determine the effect of crisaborole treatment on sleep disruption in patients with AD, the distribution of responses from patients aged 2 to < 16 years in CORE 1 and CORE 2 to item 9 of the CDLQI at baseline and day 29 is presented in Fig. 1. At baseline, 72.4% of crisaborole-treated patients and 71.2% of vehicle-treated patients reported that their sleep had been affected by their skin problem in the previous week (p = 0.192). At day 29, a significantly lower proportion of crisaborole-treated patients than vehicle-treated patients (48.5% versus 57.7%; p = 0.001) reported that their sleep had been affected by their skin problem in the previous week.

To assess the impact of a patient's AD on the sleep of the patient's family, the distribution of responses from families of patients aged 2 to < 18 years in CORE 1 and CORE 2 to item 3 of the DFI at baseline and day 29 is shown in



Fig. 1 Responses from pediatric patients (aged 2 to < 16 years) in CORE 1 and CORE 2 to CDLQI sleep item 9: "Over the last week, how much has your sleep been affected by your skin problem?" *p* values were derived from a Wilcoxon rank sum test with factor of treatment group. No missing data were imputed for these calculations. *CDLQI* Children's Dermatology Life Quality Index



**Fig. 2** Responses from families of pediatric patients (aged 2 to < 18 years) in CORE 1 and CORE 2 to DFI sleep item 3: 'Over the last week, how much effect has your child having eczema had on the sleep of others in the family?' *p* values derived from a Wilcoxon rank sum test with factor of treatment group. No missing data were imputed for these calculations. *DFI* Dermatitis Family Impact



**Fig. 3** Responses from families of pediatric patients (aged 2 to < 18 years) in CORE 1 and CORE 2 to DFI sleep item 7: 'Over the last week, how much effect has your child having eczema had on causing tiredness or exhaustion in your child's parents/carers?' *p* values derived from a Wilcoxon rank sum test with factor of treatment group. No missing data were imputed for these calculations. *DFI* Dermatitis Family Impact



**Fig. 4** Responses in CARE 1 to proxy POEM sleep item 2: "Over the last week, on how many nights has your child's sleep been disturbed because of eczema?" The full analysis set of patients included any patient receiving  $\geq 1$  dose of crisaborole. *POEM* Patient-Oriented Eczema Measure

Fig. 2. At baseline, 58.3% and 55.2% of families of crisaborole- and vehicle-treated patients, respectively, reported that the sleep of other family members had been affected by their child's AD in the previous week (p = 0.146). The proportion of families reporting that the sleep of other family members had been affected by their child's AD in the previous week was significantly lower with crisaborole than with vehicle at day 29 (35.8% versus 43.1%, p = 0.015).

The DFI also included another measure to capture the sleep impact on families of patients with AD (DFI item 7). The distribution of responses from families of patients aged 2 to < 18 years in CORE 1 and CORE 2 to item 7 of the DFI at baseline and day 29 is presented in Fig. 3. At baseline, 52.5% and 53.2% of families of crisaborole- and vehicle-treated patients, respectively, reported that their child's AD affected the parents or caregivers' feeling tired or exhausted in the previous week (p = 0.539). The proportion of families reporting that their child's AD had affected the parents or caregivers' feeling tired or exhausted in the previous week was significantly lower with crisaborole than with vehicle at day 29 (32.8% versus 39.7%, p = 0.019).

The distribution of responses from parents or caregivers of crisaborole-treated patients in CARE 1 to the proxy POEM at baseline, day 8, day 15, and day 29 is shown in Fig. 4. At baseline, 60.7% of respondents reported that their child's sleep had been disturbed due to the child's AD one or more nights during the previous week. At days 8, 15, and 29 of crisaborole treatment, that proportion was reduced to 40.3%, 32.9%, and 28.5%, respectively. No sleep-related adverse events were reported in CORE 1, CORE 2, or CARE 1.

### DISCUSSION

Despite the documented importance of sleep disruption on the negative QoL impact associated with AD, data on the effects of topical AD treatments on sleep outcomes in this patient population are limited [23-25]. This post hoc analysis examined how crisaborole treatment affected sleep outcomes for pediatric patients and their families across three clinical trials assessing the safety and efficacy of crisaborole in mild-to-moderate AD. At day 29 of CORE 1 and CORE 2, a statistically significant proportion of crisaborole-treated patients aged 2 to < 16 years reported experiencing less sleep disturbance due to their skin problem over the previous week, relative to vehicle-treated patients. Also at day 29, the families of crisaborole-treated patients aged 2 to < 18 years reported statistically significantly less disturbance to the sleep of family members because of their child's eczema, relative to the families of vehicle-treated children. Finally, the families of crisaborole-treated patients aged 2 to < 18 years in CORE 1 and CORE 2 reported statistically significantly less impact on parent or caregiver exhaustion and tiredness because of their child's eczema. At days 8, 15, and 29 of CARE 1, there was an increase in the number of respondents reporting that their child's sleep had not been disturbed by the child's eczema. The increase was observed as early as day 8 of crisaborole treatment. Although crisaborole improved sleep outcomes relative to vehicle in CORE 1 and CORE 2, sleep disturbance was noted in 48.5% of crisaborole-treated patients and about a third of families at day 29, thereby indicating the persistent impact of AD on sleep.

In addition to sleep improvement, crisaborole has been reported to improve other aspects of QoL. In an analysis by Simpson et al., patients treated with crisaborole in CORE 1 and CORE 2 reported significantly improved QoL compared with vehicle across the domains of pruritus, self-consciousness, and sexual difficulties, as based on the CDLQI (patients aged 2-15 years) and DLQI (patients aged > 16 years) questionnaires [19]. Additionally, most patients reported that AD had had a "moderate effect" on QoL at baseline or worse. By day 29, a numerically greater proportion of crisaboroletreated patients reported that AD had had a "small effect" to "no effect" on QoL than reported by vehicle-treated patients (71.8% versus 65.5%) [19]. These results support the findings of the current analysis in that crisaborole treatment can result in significant QoL improvements relative to vehicle. Also. improvements in pruritus were seen across all timepoints during the CORE 1, CORE 2, and CARE 1 studies [17, 19].

The various manifestations of sleep disruption in patients with AD are still under investigation. Studies have previously found that children with AD display both increased sleep onset latency and waking after sleep, relative to children without AD [26-28]. However, other reports show there is no difference between sleep onset in children with or without AD [29, 30]. Other abnormalities in polysomnography parameters (e.g., sleep efficiency, sleep-stage architecture, arousal, and limb movements) have been noted in children with AD, relative to published normative values [28]. There is a clear need for validated tools that can accurately reflect the impact of AD on sleep in this patient population. Beyond the use of the CDLQI and POEM in future clinical trials with pediatric AD patient populations, newer tools under development, such as the Patient-Reported Outcomes Measurement Information System, may be able to provide higher-quality sleep data in these patients [31].

Because these crisaborole studies were not specifically designed to assess the effects of crisaborole on sleep outcomes, there are limitations to the analyses. Sleep outcomes were assessed with patient- or proxy-reported measures, and no objective measurements of sleep quality, such as polysomnography or actigraphy, were obtained. There may also be differences in patient and parent, or proxy, recall of AD's sleep impacts. In CORE 1 and CORE 2, the CDLQI was used to evaluate patients aged 2 to < 16 years; however, the questionnaire has been validated only for patients aged > 4 years. Finally, the analyses presented in this manuscript were conducted post hoc, and longer-term studies on sleep outcomes in patients with AD are needed.

### CONCLUSION

After twice-daily use for 28 days to treat mild-tomoderate AD in pediatric patients, crisaborole improved sleep outcomes in both the patients and their families. Future trials are warranted to investigate the improvements in sleep associated with PDE4 inhibition in AD.

### ACKNOWLEDGEMENTS

*Funding.* This study was sponsored by Pfizer Inc. Pfizer Inc. was involved in design of the study, data collection and interpretation, writing of this manuscript, and the decision to submit for publication. Pfizer also funded the journal's Rapid Service Fees.

*Medical Writing, Editorial, and Other Assistance.* Editorial and medical writing support under the guidance of authors, including the initial drafting of this manuscript, was provided by Christopher Goodwin, PhD, at ApotheCom, San Francisco, CA, USA, and was funded by Pfizer Inc., New York, NY, in accordance with Good Publication Practice (GPP3) guidelines (*Ann Intern Med.* 2015;163:461–464).

*Authorship.* All named authors meet the International Committee of Medical Journal Editors (ICMJE) criteria for authorship for this article, take responsibility for the integrity of

the work as a whole, and have given their approval for this version to be published.

*Author Contributions.* John L. Werth, Daniela E. Myers, Daniela Graham, Alexander Agyei Marfo, and Liza Takiya contributed to study concept and design. Joseph Fowler, Jeffrey Sugarman, and Lawrence Sher acquired data. Chuanbo Zang conducted the statistical analyses of the data.

*Compliance With Ethics Guidelines.* The institutional review board at each study site approved each study's protocol, and written informed consent was provided by parents or legal guardians. The studies were conducted in accordance with the protocol, local legal and regulatory requirements, and the general principles set forth in the International Ethical Guidelines for Biomedical Research Involving Human Subjects, the International Conference on Harmonisation Guideline for Good Clinical Practice, and the Declaration of Helsinki.

**Disclosures.** Joseph Fowler has no relevant disclosures. Jeffrey Sugarman has received speaking and consulting fees from Pfizer Inc. Lawrence Sher has been a consultant and speaker for Pfizer Inc., Sanofi, Glenmark, and Regeneron. Chuanbo Zang, John L. Werth, Daniela E. Myers, Daniela Graham, Alexander Agyei Marfo, and Liza Takiya are employees of and stockholders in Pfizer Inc.

**Data Availability.** The datasets generated during and/or analyzed during the current study are available from the corresponding author on reasonable request. The datasets generated during and/or analyzed during the current study are not publicly available. Upon request, and subject to review, Pfizer will provide the data that support the findings of this study. Subject to certain criteria, conditions and exceptions, Pfizer may also provide access to the related individual deidentified participant data. See https://www. pfizer.com/science/clinical-trials/trial-data-andresults for more information.

**Open Access.** This article is licensed under a Creative Commons Attribution-NonCommercial

4.0 International License, which permits any noncommercial use, sharing, adaptation, distribution and reproduction in any medium or format, as long as you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons licence, and indicate if changes were made. The images or other third party material in this article are included in the article's Creative Commons licence, unless indicated otherwise in a credit line to the material. If material is not included in the article's Creative Commons licence and your intended use is not permitted by statutory regulation or exceeds the permitted use, you will need to obtain permission directly from the copyright holder. To view a copy of this licence, visit http://creativecommons.org/ licenses/by-nc/4.0/.

### REFERENCES

- 1. Carroll CL, Balkrishnan R, Feldman SR, Fleischer AB Jr, Manuel JC. The burden of atopic dermatitis: impact on the patient, family, and society. Pediatr Dermatol. 2005;22(3):192–9.
- Yang EJ, Sekhon S, Sanchez IM, Beck KM, Bhutani T. Recent developments in atopic dermatitis. Pediatrics. 2018;142(4): e20181102.
- Fishbein AB, Cheng BT, Tilley CC, et al. Sleep disturbance in school-aged children with atopic dermatitis: prevalence and severity in a cross-sectional sample. J Allergy Clin Immunol Pract. 2021;9(8): 3120-9.e3.
- Drucker AM, Wang AR, Li WQ, Sevetson E, Block JK, Qureshi AA. The burden of atopic dermatitis: summary of a report for the National Eczema Association. J Investig Dermatol. 2017;137(1):26–30.
- Silverberg JI, Simpson EL. Association between severe eczema in children and multiple comorbid conditions and increased healthcare utilization. Pediatr Allergy Immunol. 2013;24(5):476–86.
- 6. Tracy A, Bhatti S, Eichenfield LF. Update on pediatric atopic dermatitis. Cutis. 2020;106(3):143–6.
- 7. Ramirez FD, Chen S, Langan SM, et al. Association of atopic dermatitis with sleep quality in children. JAMA Pediatr. 2019;173(5): e190025.

- Bridgman AC, Block JK, Drucker AM. The multidimensional burden of atopic dermatitis: an update. Ann Allergy Asthma Immunol. 2018;120(6):603–6.
- Wan J, Margolis DJ, Mitra N, Hoffstad OJ, Takeshita J. Racial and ethnic differences in atopic dermatitisrelated school absences among us children. JAMA Dermatol. 2019;155(8):973–5.
- 10. Kim SY, Kim MS, Park B, Kim JH, Choi HG. Allergic rhinitis, atopic dermatitis, and asthma are associated with differences in school performance among Korean adolescents. PLoS ONE. 2017;12(2): e0171394.
- 11. Magin P, Adams J, Heading G, Pond D, Smith W. Experiences of appearance-related teasing and bullying in skin diseases and their psychological sequelae: results of a qualitative study. Scand J Caring Sci. 2008;22(3):430–6.
- 12. Drucker AM. Atopic dermatitis: Burden of illness, quality of life, and associated complications. Allergy Asthma Proc. 2017;38(1):3–8.
- 13. Staquis [prescribing information]. Beirut, Lebanon: Pfizer Labs; 2020.
- 14. Eucrisa [prescribing information]. New York, NY: Pfizer Labs; 2020.
- 15. Paller AS, Tom WL, Lebwohl MG, et al. Efficacy and safety of crisaborole ointment, a novel, nonsteroidal phosphodiesterase 4 (PDE4) inhibitor for the topical treatment of atopic dermatitis (AD) in children and adults. J Am Acad Dermatol. 2016;75(3):494-503.e6.
- 16. Yosipovitch G, Simpson EL, Tan H, et al. Effect of crisaborole topical ointment, 2%, on atopic dermatitis-associated pruritus: an extended analysis of 2 phase 3 clinical trials. Itch. 2018;3: e12.
- 17. Schlessinger J, Shepard JS, Gower R, et al. Safety, effectiveness, and pharmacokinetics of crisaborole in infants aged 3 to < 24 months with mild-to-moderate atopic dermatitis: a phase IV open-label study (CrisADe CARE 1). Am J Clin Dermatol. 2020;21(2):275–84.
- 18. Hanifin JM, Rajka G. Diagnostic features of atopic dermatitis. Acta Derm Venereol. 1980;60(92):44–7.
- 19. Simpson EL, Paller AS, Boguniewicz M, et al. Crisaborole ointment improves quality of life of patients with mild to moderate atopic dermatitis and their families. Dermatol Ther (Heidelb). 2018;8(4):605–19.
- 20. Lewis-Jones MS, Finlay AY. The Children's Dermatology Life Quality Index (CDLQI): initial

validation and practical use. Br J Dermatol. 1995;132(6):942–9.

- 21. Lawson V, Lewis-Jones MS, Finlay AY, Reid P, Owens RG. The family impact of childhood atopic dermatitis: the Dermatitis Family Impact Questionnaire. Br J Dermatol. 1998;138(1):107–13.
- 22. Charman CR, Venn AJ, Williams HC. The patientoriented eczema measure: development and initial validation of a new tool for measuring atopic eczema severity from the patients' perspective. Arch Dermatol. 2004;140(12):1513–9.
- Leo HL, Bender BG, Leung SB, Tran ZV, Leung DY. Effect of pimecrolimus cream 1% on skin condition and sleep disturbance in children with atopic dermatitis. J Allergy Clin Immunol. 2004;114(3): 691–3.
- 24. Pacor ML, Di Lorenzo G, Martinelli N, Mansueto P, Rini GB, Corrocher R. Comparing tacrolimus ointment and oral cyclosporine in adult patients affected by atopic dermatitis: a randomized study. Clin Exp Allergy. 2004;34(4):639–45.
- 25. Doss N, Kamoun MR, Dubertret L, et al. Efficacy of tacrolimus 0.03% ointment as second-line treatment for children with moderate-to-severe atopic dermatitis: evidence from a randomized, doubleblind non-inferiority trial vs. fluticasone 0.005% ointment. Pediatr Allergy Immunol. 2010;21(2 Pt 1):321–9.

- 26. Chang YS, Chou YT, Lee JH, et al. Atopic dermatitis, melatonin, and sleep disturbance. Pediatrics. 2014;134(2):e397-405.
- 27. Chang YS, Chiang BL. Mechanism of sleep disturbance in children with atopic dermatitis and the role of the circadian rhythm and melatonin. Int J Mol Sci. 2016;17(4):462.
- 28. Treister AD, Stefek H, Grimaldi D, et al. Sleep and limb movement characteristics of children with atopic dermatitis coincidentally undergoing clinical polysomnography. J Clin Sleep Med. 2019;15(8): 1107–13.
- 29. Bender BG, Leung SB, Leung DY. Actigraphy assessment of sleep disturbance in patients with atopic dermatitis: an objective life quality measure. J Allergy Clin Immunol. 2003;111(3):598–602.
- Fishbein AB, Mueller K, Kruse L, et al. Sleep disturbance in children with moderate/severe atopic dermatitis: a case-control study. J Am Acad Dermatol. 2018;78(2):336–41.
- Fishbein AB, Lor J, Penedo FJ, Forrest CB, Griffith JW, Paller AS. Patient-reported outcomes for measuring sleep disturbance in pediatric atopic dermatitis: cross sectional study of PROMIS pediatric sleep measures and actigraphy. J Am Acad Dermatol. 2020. https://doi.org/10.1016/j.jaad.2020.05. 138.